Author's response to reviews

Title: Benign mesenteric lipodystrophy presenting as low abdominal pain is a surgical minefield: A case report

Authors:

Jonathan Richard Rees (j.r.e.rees@btinternet.com)
Phillip Burgess (phillip.burgess@gwh.nhs.uk)

Version: 2  Date: 11 February 2010

Author's response to reviews: see over
Dear Sir/Madam,

I have made the following revisions to the manuscript:-

**Major**

Title amended to include a case report

Abstract from manuscript copied into online system so that documentation is consistent

Ethnicity and nationality of patient included

Contributing authors statement amended to read “both authors read and approved the final manuscript”

**Minor**

Tracking of changes system switched off

Title page amended to remove Received and accepted dates

Yours sincerely

Jonathan Rees